Author’s response to reviews

Title: No increased risk of glucose metabolism disorders in adults with growth hormone deficiency undergoing long-term treatment with biosimilar somatropin (Omnitrope®): data from an observational, longitudinal study

Authors:

Paolo Beck-Peccoz (paolo.beckpeccoz@unimi.it)
Charlotte Höybye (charlotte.hoybye@sll.se)
Robert Murray (robertmurray@nhs.net)
Suat Simsek (S.Simsek@nwz.nl)
Markus Zabransky (markus.zabransky@sandoz.com)
Hichem Zouater (hichem.zouater@sandoz.com)
Günter Stalla (guenter.stalla@medicover.de)

Version: 3 Date: 19 Nov 2019

Author’s response to reviews:

Editor Comments:

1. Author Affiliations
   Please ensure that the affiliations of Hichem Zouater includes Sandoz i.e. as done for Markus Zabransky, on the title page, as per the Competing Interests.

   RESPONSE: On the manuscript title page, it is indicated that affiliation number 5 applies to both Markus Zabransky and Hichem Zouater (5 Sandoz Biopharmaceutical, c/o HEXAL AG, Industriestr. 25, D-83607 Holzkirchen Holzkirchen, Germany)

2. Clean manuscript
   At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   RESPONSE: Clean versions of all files have been provided.